Author's response to reviews

Title: Sequence analysis of genes mediating extended-spectrum beta-lactamase (ESBL) production in isolates of Enterobacteriaceae in a Lagos Teaching Hospital, Nigeria.

Authors:

Muhabat A Raji (adeola68@gmail.com)
Wafaa - Jamal (wjamal@hsc.edu.kw)
Omoh F Ojemeh (ojemhen@gmail.com)
Vincent O Rotimi (vincent@hsc.edu.kw)

Version: 2 Date: 26 August 2014

Author's response to reviews: see over
Cover Letter

To: Editor, BMC Infectious Diseases.

From: Dr Muhabat Adeola Raji, Department of Medical Microbiology, Lagos State University, College of Medicine, Ikeja, Nigeria.

Re: Changes made to the submitted manuscript

The following changes have been effected in the body of the submitted manuscript.

1. A statement has been made on competing interests by the authors.

2. The individual author’s contribution has been included.

3. Acknowledgements have been included.

4. A statement on ethical approval has been made.

5. The emails of all authors have been included.

6. The conclusion and discussion portions of the manuscript are now separate.

Thank you.

Yours Sincerely,

Dr. Muhabat Adeola RAJI.